Author's response to reviews

Title: Effect of lifestyle intervention for (pre)diabetics in real-world primary care: propensity score analysis.

Authors:

Joris J Linmans (joris.linmans@maastrichtuniversity.nl)
Mark G Spigt (m.spigt@maastrichtuniversity.nl)
Linda Deneer (l.deneer@maastrichtuniversity.nl)
Annelies EM Lucas (a.lucas@sge.nl)
Marlies de Bakker (m.debakker@sge.nl)
Luc G Gidding (luc.gidding@maastrichtuniversity.nl)
Rik Linssen (rik.linssen@maastrichtuniversity.nl)
J André Knottnerus (andre.knottnerus@maastrichtuniversity.nl)

Version: 2 Date: 12 May 2011

Author's response to reviews:

Maastricht, 12-05-2011

Dear Editor,

We would like to submit our manuscript titled: “Effect of lifestyle intervention for (pre)diabetics in real-world primary care: propensity score analysis” to BMC Family Practice.

We think that our manuscript fits perfectly within the scope of BMC Family Practice. We evaluated the effects of a nationwide lifestyle programme for patients with (pre)diabetes in real-world primary care. The results have important implications for healthcare providers, researchers and policymakers in many different areas of medicine.

The corresponding author takes full responsibility for the contents of the paper. We hope you will consider our manuscript for publication in your journal.

Also on behalf of the other authors,

Yours sincerely,

Joris Linmans, MD
Corresponding author.

Joris J Linmans
Maastricht University, CAPHRI, Department of General Practice
P.O. Box 616
6200 MD, Maastricht
The Netherlands
joris.linmans@maastrichtuniversity.nl